Author's response to reviews

Title: The efficacy and safety of insulin-sensitizing drugs in HIV-associated lipodystrophy syndrome: a meta-analysis of randomized trials

Authors:

Siddharth H Sheth (siddharth.h.sheth@dartmouth.edu)
Robin J Larson (robin.j.larson@dartmouth.edu)

Version: 4 Date: 28 April 2010

Author's response to reviews: see over
To the BMC ID Editors:

We thank you for accepting our manuscript, “The efficacy and safety of insulin-sensitizing drugs in HIV-associated lipodystrophy syndrome: a meta-analysis of randomized trials”. As requested by the Editors, we made the major and minor revisions requested by the editorial staff. In addition to the revised manuscript, we have also re-submitted the tables, figures, and additional files. Below is a point-by-point response to revisions requested by the editorial staff.

Major Revisions:

1. An additional page was created in the manuscript citing additional files. In total, there are three additional files. These files have been submitted.

Minor Revisions

1. Author’s qualifications have been removed from the manuscript and submissions system.

2. Country details have been added to the affiliations.

3. The disclaimer has been removed from the title page and a newly created Acknowledgement section has been added.

4. All highlighting and tracked changes have been removed.

5. The introduction section has been renamed as “Background”.

We thank you again for accepting our manuscript. Please feel free to contact us with any additional questions.

Sincerely,

Siddharth Sheth, MPH
Robin Larson, MD MPH